Author's response to reviews

Title: Analysis of folylpoly-gamma-glutamate synthetase gene expression in human B-precursor ALL and T-lineage ALL cells

Authors:

Guy J Leclerc (leclergj@musc.edu)
Gilles M Leclerc (leclergm@musc.edu)
Ting Ting Hsieh-Kinser (kinser@musc.edu)
Julio C Barredo (barredjc@musc.edu)

Version: 6  Date: 11 May 2006

Author's response to reviews:

May, 11th, 2006

Dr. Denys N Wheatley
Editor in chief
Department of Cell Pathology,
University of Aberdeen,
581 King Street,
Aberdeen AB24 5UA,
Scotland, UK.
Tel: 44-1224-274172
Fax: 44-1224-274179

Dear Dr. Denys N Wheatley,

Dr. Leclerc and I thank you and the Referees for the acceptance of our manuscript entitled "Analysis of folylpoly--glutamate synthetase gene expression in human B-precursor ALL and T-lineage ALL cells" for publication in BMC Cancer (MS #: 2223099688970260). As requested, we cited website addresses and removed them from the manuscript text. In addition, we moved the figure legends and Table 1 to after the Reference list.

Thank you again for your help and consideration.

Sincerely,

Julio C. Barredo, M.D.
Professor of Pediatrics and Biochemistry
And Molecular Biology
Director, Pediatric Hematology-Oncology
Medical University of South Carolina
Director, Children's Cancer Programs
MUSC Hollings Cancer Center
135 Rutledge Ave.
Charleston, SC 29425
843-792-2957
843-792-8912 (fax)